DRAFT_2
EO-1 Weekly Status Week of July 31, 2013 – Aug 06, 2013
Day of Year 212 - 218
Mission Day 4696 – 4702
Earth Observing One (EO-1) - General
Instruments

· Scheduled 131 science Data Collection Events (DCEs) 

· Received all images

· Performed Instrument decontamination cycle 

· Conducted HSI deicing from 217/00:25z to 217/15:15z

· Conducted ALI outgassing from 217/00:35z to 217/15:25z

· Performed Instrument calibration

· Conducted solar array characterization at 217/10:53z
· Conducted HSI solar calibration at 217/20:46z
Technology Activities

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintaining nominal spacecraft state of health

· Designing the added steps to add goal upload automation to 32K TDRS supports

· Monitoring voltage and differential voltage

· Working with FSW personnel to change TSM 54 (Battery Differential Voltage) limits.

· Limit Levels have been decided upon and will coincide with TRMM who has the same type of battery as EO-1

· Researching why ITPS won’t produce reports on some already ingested data 

· Ingesting remaining data from year 2010

· Investigating the possibility of reducing the slew and settling time of the first image of the day.

· Reviewing UDAPs from years 2004 and 2005 to check slew times

· Reviewing x-band power constraints for image and pass scheduling using ASPEN.

Mission Planning

· Removing the AC commands from the Lunar Calibrations entirely (before the AC2/3 scan was removed but AC1 was still being taken on scan 4)

· Working with mission scientist to create Hyperion only lunar calibration and adjust pointing slightly

· Working with ASPEN developers to configure various Lunar Calibration sequences 

· Reviewed single scan lunar calibration UDAPS from ASPEN

· Reviewed the 4 scan ALI centered calibration UDAP from ASPEN

· Requested a 4 scan HSI centered lunar calibration from the ASPEN developer

· Investigating the steps needed to perform an HSI solar cal from an ATS load

· Working with ASPEN developer on solar calibration

· Working with the ASPEN developer to allow the ingestion of 32K TDRS supports into ASPEN

· The ASPEN developer will add RTS45 related commands to the ASPEN system to allow for TDRS 32K UDAP generation.

· Looking into what would be needed for manual testing of reducing the slew and settling time of the first image of the day

Flight Dynamics 

· Working with mission scientist to create Hyperion lunar calibration and slightly adjust pointing

· Working with ASPEN developers to rebuild lunar calibration

· Testing out various single scan and 4 scan HSI and ALI targeted lunar calibrations

· Working with developers on small discrepancies

· Researching if range data from ground stations would improve OD results

· The EO-1 spacecraft can support tone ranging

· Looking up information about major and minor tones

· Working with the conjunction assessment group on small discrepancies

· Looking into adding Range data to our Range-Rate data that we currently use for tracking data, this should improve results

Central File Hub

· Changes to sync_hubs to better streamline and retain metadata on files have been put in place

· Testing is complete and the new version has been implemented.

· Phase one tests are ready for outside entity delivery.
· White Sands incoming file testing is complete and files are only delivered to the hubs.

· JPL incoming file testing is complete and files are only delivered to the hubs.

· PF incoming file testing is complete and files are only delivered to the hubs. 

· SGS and WGS tracking data are waiting for a break in WALLOPS activities to begin testing.

· Phase two scripts are designed, created, and in-house tested. 
· Phase 2 scripts for MP and Realtime are being tested and prepared to put into place
System Administration

· Performing EO-1 mission firewall cleanup.

· Removed access from USN to the Flight Dynamics computers since the migration to the Central File Hubs is complete. Also removed from Windows firewall.

· Removed multiple unnecessary rules that the logs show weren’t used in over a year. Worked with IONet/GCC personnel to gather information to ensure their removal wouldn’t impact operations.

· Added a rule to allow the firewall to send logs to the regional logserver without the traffic being logged. By default all outbound traffic originating from the firewall is logged.

· Completed sync_hubs.pl script update on the Central File Hubs. Changes described below.

· Only the backup file hub automatically moves duplicate/already delivered files into the migrated folder. The primary redelivers the files in case the contents have been updated.

· The primary hub only syncs files from the backup hub that are don’t exist on the primary. This makes the script more efficient and keeps the modified date accurate.

· Configured key-based SSH authentication between accounts on the primary and backup hubs that are necessary for the automated file listing and comparison described above. 

· Completed SNAS certificate renewal for the 3 FOT accounts. Certificates were generated on the SNAS site and installed on the Flight Dynamics computers.

· Completed the periodic reboot of the backup string of operational computers. A chkdsk/fsck is performed on each system during this routine maintenance.

· Completed routine root/admin password changes on all operational, Flight Software Lab, and CNE systems.

· Updated the spreadsheet that acts as a summary for EO-1’s IONet firewall rules. Recent rule changes, new expiration dates, and changes related to the new MSGRS tool were recorded.

· Completed the required “Managing System Security in Windows Server 2003” SATERN course.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, regional logserver and MOC CNE computer.

· Created weekly tape.

· Delivered December monthly tape to MSL.

GROUND AND SPACE NETWORK

Station Downtimes 


· HGS is down for at least the next 2 months

Operational Discrepancies

· None

Number of Real-Time Supports


[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

31-Jul-13 212 7 3 0 0 10

1-Aug-13 213 7 1 1 0 9

2-Aug-13 214 7 3 1 0 11

3-Aug-13 215 5 5 0 0 10

4-Aug-13 216 5 4 0 0 9

5-Aug-13 217 1 6 0 0 7

6-Aug-13 218 5 2 1 0 8

37 24 3 0 64 Weekly Totals


UPCOMING EVENTS

· Continuation of Phase 1 central file hub from FD, Start of Phase 2 for MP and Realtime 

· Testing of ASPEN lunar calibration UDAPS

The total number of ALI scenes in the level-0 archive: 69914

The total number of HYP scenes in the level-0 archive: 69646

The number of ALI level-0 scenes ingested in the past 7 days is 137

The number of HYP level-0 scenes ingested in the past 7 days is 138
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		31-Jul-13		212		7		3		0		0		10

		1-Aug-13		213		7		1		1		0		9

		2-Aug-13		214		7		3		1		0		11

		3-Aug-13		215		5		5		0		0		10

		4-Aug-13		216		5		4		0		0		9

		5-Aug-13		217		1		6		0		0		7

		6-Aug-13		218		5		2		1		0		8

		Weekly Totals				37		24		3		0		64
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